Author's response to reviews

Title: An adult patient presenting with appendix perforation in appendix duplication: a case report

Authors:

Emel Canbay (dremelcanbay@gmail.com)
Emel Akman (emelakman1@hotmail.com)

Version: 9 Date: 7 November 2010

Author's response to reviews: see over
November 7, 2010

Dear Mr Ed Bautista,

Thank you very much for your letter. We have revised the our case report. We are sending the manuscript again with required changes such as 1) the ethnicity of the patient in the abstract section of the manuscript was given, 2) the references were given in brackets. 3) additional/supplementary files were removed and 4) the figure was cropped to minimize white spaces 4) table format was changed to landscape and drawn with black to visualize the vertical lines.

We are looking forward to hearing from you soon.

Sincerely yours

Emel Canbay MD,PhD

Corresponding author